Contents lists available at ScienceDirect

Ecological Economics

journal homepage: www.elsevier.com/locate/ecolecon



CrossMark

Adán L. Martínez-Cruz^{a,c,*}, Miriam Juárez-Torres^b, Santiago Guerrero^{b,d}

^a Department of Economics, Centro de Investigación y Docencia Económicas (CIDE), Circuito Tecnopolo Norte 117, Aguascalientes, ACS 20313, Mexico

^b Banco de México, Avenida Cinco de Mayo 18, Centro, Cuauhtémoc, 06059 Ciudad de México, D.F., Mexico

^c Centre for Energy Policy and Economics (CEPE), ETH-Zurich, ZUE E8, Zurichbergstrasse 18, 8092 Zurich, Switzerland

^d DINAMA, Galicia 1133 Montevideo, Uruguay, 11100; CINVE, Avenida Uruguay 124, Montevideo, Uruguay, 11100

ARTICLE INFO

Article history: Received 18 December 2015 Received in revised form 15 December 2016 Accepted 2 March 2017 Available online xxxx

JEL Classification: C80 C83 Q15 Q54

Keywords: Expert elicitation Expected impact Climate change Irrigation Local social-ecological systems Double-bounded dichotomous question Bolivia

ABSTRACT

Assessment of the expected impacts from climate change is an essential input for agencies engaged in fostering adaptation of local social-ecological systems. However, data is seldom available at the required scale. This study overcomes this hurdle by gathering data via an expert elicitation protocol. We report experts' judgements about two topics: i) the impacts from climate change on crop yields in three communities located in the Bolivian Altiplano; and ii) the effectiveness of specific irrigation techniques in mitigating the impacts from climate change in the communities under study. Our gathered data allow us to document heterogeneity of expected impacts across communities – with one community expected to experience an increase in yields under wet climate change scenarios. Experts judge irrigation as an effective mitigation tool under most of the dry climate change scenarios presented to them. We believe that our data collection strategy represents a promising decision support tool for a wide range of public policy issues. Particularly, when monetary and time constraints converge with the lack of scientific information. In addition, the information gathered with this methodology can be incorporated into participatory methodologies gathering information from local social-ecological systems.

© 2017 Elsevier B.V. All rights reserved.

1. Introduction

Climate change has been characterized as local in its causes and global in its impacts (Shaw et al., 2009). Unfortunately, documentation of the impacts from climate change has mostly focused on the global scale. This focus is not surprising given that data at local scale are frequently of poor quality and/or seldom available.

There is a pressing need to collect information at local scale. This is so because if the causes of climate change are local, the strategies to build resilience may be found at the local scale as well. And this has been the case. For instance, Mistry et al. (2016) has successfully implemented a participatory strategy that helps communities in the identification of local best practices for social-ecological sustainability. Also, local information is essential in the documentation of the expected heterogeneity of climate change impacts (Vincent, 2007) –allowing, for instance, the identification of local social-ecological systems¹ potentially benefiting from climate change.

National and international agencies have recently increased their efforts to gather data at local scales. For instance, the Inter-American Development Bank has stated its intention of incorporating measures of local adaptive capacity in evaluations of community projects involving



Analysis



[☆] This project has been carried out with resources from the Inter-American Development Bank –through its Climate Change and Sustainability Division. Authors are the sole responsible of opinions expressed in this document. We want to thank the experts in Agronomy and Climate Change that kindly participated in this study. We are thankful to Carlos Rodriguez and Corina Apaza for their help through all the stages of the expert elicitation protocol. We also thank four reviewers whose comments have greatly improved this manuscript.

^{*} Corresponding author at: Centre for Energy Policy and Economics (CEPE), ETH-Zurich, ZUE E8, Zurichbergstrasse 18, 8092 Zurich, Switzerland.

E-mail addresses: madan@ethz.ch (A.L. Martínez-Cruz), mjuarez@banxico.org.mx (M. Juárez-Torres), sguerreroe@gmail.com (S. Guerrero).

¹ The term *social-ecological system* is used to emphasize "the integrated concept of humans in nature and to stress that the delineation between social and ecological systems is artificial and arbitrary" (Folke et al., 2005, p. 443).

water and sanitation, technology innovation, and irrigation (IADB, 2013). Also, the Paris Agreement, signed in December 2015 in the COP21 meeting, is expected to increase the demand for local information given that the developed countries have committed to increase funds to avert local damages provoked by climate change (UNFCCC, 2015, p. 26).

This study explores the feasibility of collecting local data via a protocol that gathers the opinion from experts on how climate change will impact local social-ecological systems. Treating expert judgement as a type of scientific data that fits into an empirical research strategy is the departure point of the expert elicitation literature (Bolger and Rowe, 2015; Cooke and Goossens, 2000). For instance, Vaissière et al. (2013) incorporate the opinion of experts in a methodology that compares maintenance costs associated with compensating the damage of ecosystem services. They request experts to quantify the contribution of habitats and species to the production of an ecosystem service.

This study delivers two pieces of information. The first piece reflects the experts' judgements about the climate change impacts on crop yields in communities located in the Bolivian Altiplano. We have selected this region because it illustrates that the evidence about effects from climate change in the Altiplano region has not translated into a data gathering strategy that may support the design of policies tackling the expected impacts to the local social-ecological systems (see World Bank, 2010).

Informing policy makers about the expected impacts from climate change is not enough. They also need information about the effectiveness of potential mitigation actions. Thus, the second piece of information reported by this study captures experts' judgements on whether specific irrigation techniques are expected to be effective in mitigating the effects from climate change. We deem irrigation as a potential mitigation tool because it provides protection against precipitation volatility associated with climate change (Lybbert and Sumner, 2012; Vermeulen et al., 2012). To gather data valuable to policy makers, our irrigation scenarios have been designed to resemble irrigation projects implemented through the Bolivian National Irrigation Program.

The rest of this paper is structured as follows. Section 2 justifies our focus on communities located in the Bolivian Altiplano. Section 3 describes our methodology. Section 4 reports the opinions of our experts. Section 5 presents a methodological discussion. Section 6 presents our conclusions.

2. Case Studies

The Bolivian Altiplano illustrates the problem of data availability regarding impacts from climate change at the local scale. On one hand, there is enough evidence to argue that climate change is already impacting the Bolivian Altiplano via extreme weather events and higher weather variability (World Bank, 2010). For instance, extreme weather events such as El Niño provokes rainfall decline in the Altiplano (Andersen and Mamani, 2009). In 2010, a more severe drought affected the area due to the shortening of the rain season (MMAAB, 2009). An unprecedented hailstorm in 2002 caused a U.S. \$70 million damage and droughts with a bigger-than-usual influence area. These events are becoming repetitive and persistent, intensifying or triggering socioeconomic phenomena such as migration from the countryside to the cities.

On the other hand, despite the documented evidence of the effects of climate change on the Bolivian Altiplano, the greatest challenge for the social-ecological systems in this region is the uncertainty about the geographical distribution of such impacts (World Bank, 2010). Due to the unique topographic conditions of the Bolivian Altiplano, researchers expect a large heterogeneity in the effects from climate change on local social-ecological systems –with chances that some systems may benefit from climate change and others may face negative impacts.

Thus, as a first step to design a strategy to face climate change, policy makers need to be informed with estimates of the expected impacts not only at the regional scale but also at the local scale –with emphasis on the identification of the local systems that may gain and the local

systems that may be damaged by climate change. However, information at a local scale is seldom available for this region (World Bank, 2010).

In this context, we implement our expert elicitation methodology in the Bolivian Altiplano. Three Altiplano communities are taken as case studies. As illustrated in Fig. 1, Suriri-Capiri, Pasacunta-Qollpacanta, and Peñas Kerani are part of, respectively, the municipality of Tiwanaku, the municipality of Calacoto, and the municipality of Batallas. The three communities are in the administrative department of La Paz –which is part of the dry Altiplano. In this sub-region, low temperatures, high variability of temperatures, low humidity, low rainfall and recurrent and long periods of drought impose severe conditions to crop and livestock production (World Bank, 2010).

The three communities under study are useful to gain insights on how effective irrigation may be at mitigating the impacts from climate change. These communities have benefited from irrigation projects financed by PRONAREC – the National Irrigation Program with a Watershed Approach (IDB, 2008). Thus, the irrigation scenarios included in our protocol closely resemble the projects financed by PRONAREC – which allows us to gain insights of public policy relevance.

3. Methodology

We collect data through an expert elicitation (EE) protocol. EE is a structured process that collects scientific and technical judgements from experts (Morgan, 2014; Bosetti et al., 2016). EE is deemed a useful piece of empirical research, particularly when other empirical data is expensive, limited or unreliable (Bolger and Rowe, 2015; James et al., 2010).

Bosetti et al. (2016) describes EE in eight steps –these steps are executed in an iterative manner and thus a practitioner may return to a specific step several times. In the first step, an EE describes the objective and mode of the elicitation. The definition of the goal includes an unambiguous description of the metric that experts are requested to use when reporting the parameter of interest. The elicitation mode ranges from face-to-face interviews to protocols that can be self-administered via a web-based platform –importantly, the selection of the elicitation mode is not without trade-offs (see Baker et al., 2014; Verdolini et al., 2015).

In the second step of an EE, the practitioner defines the type of expertise needed and identifies the individuals with such expertise. In the third step, the type of expertise and the goal of the EE are taken into consideration to decide the format of the elicitation question. In the fourth step, experts are provided with background material and trained in the rationale behind the elicitation question. This step may be carried out in advance to the implementation of the protocol or as a preliminary step during the elicitation. In the fifth step, the EE protocol is tuned by piloting parts or the entire protocol –making sure that the metric of interest is clear to the experts and that, if it is the case, the hypothetical scenarios are precisely described.

The sixth step consists in gathering the data via the elicitation protocol. The statistical analysis of the gathered data is the seventh step of an EE. The final element is the reporting of the experts' opinions (see Bosetti et al. (2016) for details about these steps).

The rest of this section provides the specifics of the EE protocol implemented in this paper.

3.1. Expert Elicitation Protocol

3.1.1. Goal

The goal of our EE protocol is twofold. First, it collects and summarizes the opinion of experts about the expected impacts from climate change on potato yields in three communities located in the Bolivian Altiplano. Second, it collects and summarizes the opinion of experts with respect to whether irrigation is an effective tool to mitigate the changes in potato yields in the communities under study.

Potato has been chosen as the crop of interest because it is the most important crop in the Bolivian Altiplano (World Bank, 2010) –both in



Fig. 1. Location of communities under study.

terms of percentage of family income coming from this crop (50%–80%), and hectares dedicated to this crop (half of the crop surface).

3.1.2. Our Experts

The selection of experts is key to our methodology. For purposes of this study, an expert is an individual that has worked at and/or studied the local social-ecological systems in the Bolivian Altiplano. This individual is expected to know the factors that impact crop yields –with emphasis on potato. Ideally, experts have developed a mental model that maps variations in precipitation and temperature into variations in potato yields. Under this description, a potential expert may work directly with communities in the Bolivian Altiplano, or may have worked with them earlier and his/her current professional activities are such that his/her information about these communities is regularly updated.

A list of 21 experts was put together in January 2014. This lists was initiated with the suggestions from Agronomists Carlos Rodriguez –Director of PRONAREC in 2014—, and Corina Apaza –hired to assist the research team in the process of designing and implementing the elicitation protocol, and recruiting the experts. Experts contacted at this stage were also invited to suggest the names of additional experts.

An email invitation was sent to all 21 experts. Eighteen of them replied positively. These 18 experts were personally visited by the leading co-author of this paper during the third week of January 2014. During these visits, experts were informed about the goals and structure of the expert elicitation protocol. These 18 experts were invited to answer the final version of the protocol during the second week of July 2014 at La Paz, Bolivia. Twelve of them were available to answer the protocol.

Table 1 describes the professional activities and expertise of the 12 experts that responded our protocol. All 12 experts are agronomists.

Table	1
-------	---

Experts participating in the elicitation protocol.

-			
	Expert	Sector	Area of expertise
	Giovanna Salinas Murillo	Government	Climate change
	Rosa Zavala	Government	Climate change
	Octavio Boris Arias Irusta	Government	Risk analysis
	Lucio R. Tito Villca	Academy/government	Risk analysis
	Israel Romero Salazar	Technician	In field
	Eliana Espejo Ticona	Technician	In field
	Victor Cortez Limachi	Technician	In field
	Manuel Chino Velasquez	Technician	In field
	Carlos Hector Escobar Ramos	Technician	In field
	Emilio Garcia Apaza	Academy	Agronomy/climate change
	Eleodoro Baldiviezo Estrada	NGO	In field
	Maria Quispe	NGO	In field

Two experts work in a governmental agency dealing with climate change challenges. Two experts work in a governmental agency dealing with risk analysis –one of these experts is an academic researcher that has published books describing the agronomic conditions of the Altiplano. The main activities of these four experts do not involve regular visits to the communities in the Altiplano. However, they did work in the area of interest earlier in their professional career.

The other eight experts were working with several communities in the Altiplano at the time they answered our protocol. Five experts are technicians directly working at the communities. One expert is an academic researcher with focus on the Altiplano communities. Two experts work for an NGO that focuses on improving productivity in the Altiplano.

3.1.3. Elicitation Mode, and Implementation

Experts' judgements were elicited in three sessions – two workshops attended by 5 and 6 experts, respectively, and one personal interview. All three sessions had an identical structure, and took place during the second week of July 2014 at La Paz, Bolivia. During the first 30 min, experts were remained of the EE protocol's goal, and the context in which the study was carried out. During the second half-hour, the facilitator explained the rationale behind the elicitation questions, and experts were faced to examples of the EE questions. During the last 2 h of the session, experts answered the EE protocol.

Experts were encouraged to exchange points of view during the first hour of the workshop. Also, they were encouraged to answer the protocol taking into consideration what had been discussed but without consulting to each other during the last 2 h of the workshop.

3.1.4. Farming Conditions, Average Weather Conditions, and Irrigation Scenarios

Experts cannot provide unconditional judgements –i.e. if an expert is asked about average yields, he/she will first want to know the conditions under which he/she is supposed to forecast the yields. Keeping the social system fixed, a non-exhaustive list of the ecological factors impacting crop yields include the type of soil, the cropping season, and the average and variation of the temperature and the precipitation. Also, the conditions of the previous cropping season matter when forecasting expected yields.

Indeed, elements of the social system matter as well because socialecological systems are complex and include reciprocal feedbacks (Folke et al., 2005; Ostrom, 2009). A non-exhaustive list of social factor impacting the crop yields include whether farming activities are the farmer's main source of income, and local governance.

Thus, the research team extensively discussed how to describe the agronomical scenario under which experts are requested to report expected yields. In this process, we received the assistance of Agronomist Corina Apaza –who has worked with and at communities in the Bolivian Altiplano. In addition, an expert answering a pilot protocol was also consulted on the adequacy of the agronomical scenario. The final agronomical scenario aims to be understandable and simple enough, and to include the most essential factors impacting yields.

Six farming factors are included in our agronomical scenario: i) the variety of potato; ii) the type of soil; iii) whether chemical fertilizer is used; iv) whether pest control measures are implemented; v) cropping season; and vi) irrigation technique.

The farming conditions have been described to the experts as follows:

Consider a community situated in the Altiplano. This community belongs to the **municipality**, **province**, in the administrative department of La Paz. Consider a farmer in this community. This farmer grows **variety of potato**, in his/her <u>private one-hectare</u> parcel which is composed primarily of **soil type**. He/she uses **irrigation technique**, chemical

Table 2

Farming conditions and irrigation scenarios in communities under study.

Characteristics	Communities under study				
	Suriri-Capiri	Peñas Kerani			
Municipality Province Potato variety Soil type Irrigation 1 Irrigation 2	Tiwanaku Ingavi Huaycha Clay loam No irrigation Stone-lined channels	Calacoto Pacajes Luki Sandy loam No irrigation Stone-lined channels	Batallas Los Andes Huaycha Clay loam No irrigation Open channel		

<u>fertilizers</u>, and <u>no pest control strategies</u>. Preparation of lands is carried out from <u>August to September</u>. Planting is carried out from <u>mid-October</u> to mid-November. Harvest is expected from March to April.

Notice that the community of reference is not named but we do specify the municipality and province where each community is located –Table 2 lists this information. This strategy is meant to encourage experts to keep in mind factors that are similar among communities located within a municipality but vary for communities located in different municipalities –in econometric jargon, this strategy encourages the expert to control for municipality fixed effects. This type of factors cannot be exhaustively listed, and their relevance may differ depending on the expert. Examples of such factors may include the distance to Titicaca lake, the distance to the La Paz, capital of Bolivia, etc.

Table 2 lists the varieties of potato and the soil types for each community. Potato has adapted very well to the climate condition in the Altiplano –in particular, the Huaycha and Luki varieties. The Huaycha variety grows well in clay loam soil which is the most common soil in Suriri-Capiri and Peñas Kerani. The Luki variety grows well in sandy loam soil and is particularly resistant to frosts which are recurrent in Pasacunta-Qollpacanta.

Table 2 reports the irrigation scenarios considered for each community –no irrigation and a given irrigation technique. The irrigation technique at each community corresponds to the one financed by PRONAREC –water distribution through open channel in Peñas Kerani, and stone-lined channels in Pasacunta-Qollpacanta and Suriri-Capiri.

Three factors are kept fixed across the three communities: i) type of fertilizer is described as chemical fertilizer; ii) no pest control measures are implemented; and iii) the cropping season runs from August to April. These features are representative of communities in the Altiplano.

Together with the farming conditions, experts are informed about the average precipitation/temperature conditions for the 1990–2013 period. Table 3 reports these conditions for each community. The average values correspond to the observed average monthly values from October to February. Weather conditions during these months are essential for germination and development of potato in the Bolivian

Table 3

Average precipitation/temperature conditions (from 1990 to2013)^a.

Variables	Oct.	Nov.	Dec.	Jan.	Feb.
Suriri-Capiri					
Precipitation (mm)	25.60	43.00	69.40	108.90	85.20
Temperature (°C)	9.10	10.10	10.10	10.10	9.90
Pasacunta-Qollpacanta	Pasacunta-Oollpacanta				
Precipitation (mm)	20.80	24.00	80.00	123.60	88.70
Temperature (°C)	8.50	9.90	10.80	10.60	10.50
Peñas-Kerani					
Precipitation (mm)	36.10	45.70	113.40	125.20	95.90
Temperature (°C)	8.80	9.50	9.80	9.50	9.40

^a Manually downloaded from SENAMHI at http://www.senamhi.gob.bo/meteorologia/ formularioboletinmensual.php. Altiplano. Temperature and precipitation values are obtained from the closest climatological stations to the communities under study. Information has been downloaded manually from the official webpage of the Bolivian agency in charge of measuring weather conditions.

3.1.5. Climate Change Scenarios

Climate change, in this study, is characterized through four precipitation/temperature scenarios. These scenarios aim to capture weather conditions likely to hold under climate change conditions during the cropping season of interest. To better convey the type of weather conditions we aim to represent, we label the precipitation/temperature scenarios as *dry* or *wet* –although scenarios are presented unlabeled to the experts.

Two documents have been consulted to design the climate change scenarios. The first one is an *official* reference in the sense that it has been endorsed by the Bolivian government through its Ministry of Environment. This Ministry has issued official forecasts of climate change scenarios for the years 2001 to 2030, and 2071 to 2100 (MMAAB, 2009). This official reference reports that the Altiplano is expected to face an increase in temperature ranging from 0.5 °C to 1.5 °C by 2030; and annual precipitation is expected to remain the same or decrease by 15 mm.

The second consulted document is a World Bank report on the consequences of climate change (World Bank, 2010). This report suggests two climate change scenarios are plausible in 2030. A *wet* scenario would bring an increase of 1.55 °C in average temperature and a 22% increase in the annual mean precipitation. A *dry* scenario would imply an increase in temperature by 2.41 °C and a 19% decrease in precipitation.

The official scenario somehow contains the two World Bank scenarios. With respect to temperature, the official scenario assumes a value close to the maximum temperature of the wet World Bank scenario. With respect to precipitation, the range of variation in the official scenario includes the values of the dry World Bank scenario. Consequently, the scenarios designed in this study aim to capture the intersection of both the official and the World Bank reports.

Table 4 reports the climate change scenarios designed for each community under study. The four scenarios were unlabeled when presented to the experts. However, for exposition purposes, Table 4 labels them as *dry observed* (DO), *wet observed* (WO), *wet climate change* (WCC), and *dry climate change* (DCC).

On one hand, the DO and WO scenarios together capture the conditions suggested by the official scenario. The DO and WO scenarios have recently occurred –they have been chosen among the years with available measures near each community. They are labeled as dry or wet with respect to the average conditions presented in Table 3. For instance, under the DO scenario for Suriri-Capiri, precipitations from November to January are lower than in the average scenario – 22.90 mm, 18.50 mm, and 82.50 mm versus 43.00 mm, 69.40 mm, and 108.90 mm. Also, during the same months, average temperatures in the DO scenario are lower in comparison to the average scenario – 9.10 °C, 10.10 °C, and 9.80 °C versus 10.10 °C, 10.10 °C and 10.10 °C. A similar reasoning is behind the DO and WO scenarios for each community.

On the other hand, the DCC and WCC scenarios resemble the wet and dry World Bank scenarios. The values for the DCC and WCC are hypothetical –i.e. they have not occurred but can plausibly occur— and have been calculated with respect to the average scenarios listed in Table 3. The WCC scenario implies an increase in monthly average temperature of 1.7 °C, and an increase of 28% in monthly precipitation. The DCC scenario implies an increase in temperature of 2.5 °C and a reduction of 20% in precipitation.

3.1.6. Elicitation Questions

Experts' judgements are elicited through a five-step procedure – which is depicted in Fig. 2. In the first step, the agronomical scenario is presented to the experts. This scenario contains a description of the farming conditions and the average precipitation and temperature conditions (see Section 3.1.4). Experts are told to keep this agronomical scenario in mind through the elicitation procedure.

In the second step, an irrigation scenario is presented to the experts. All experts are presented to two irrigation scenarios for each community under study. In a first round of elicitation questions, the agronomical scenario is presented together with a no irrigation scenario. In a second round of elicitation questions, the same agronomical scenario is

Table 4

Precipitation and temperature conditions in climate change scenarios.

	-					
Scenario	Variables	Oct.	Nov.	Dec.	Jan.	Feb.
Suriri-Capiri						
Dry observed (1999)	Precipitation (mm)	38.80	22.90	18.50	82.50	88.50
	Temperature (°C)	8.80	9.10	10.10	9.80	9.50
Wet observed (2013)	Precipitation (mm)	20.50	10.50	129.50	127.70	92.50
	Temperature (°C)	9.10	10.60	10.20	9.50	9.70
Dry climate change	Precipitation (mm)	18.08	29.04	62.64	78.88	64.56
	Temperature (°C)	13.00	14.00	14.00	13.80	13.60
Wet climate change	Precipitation (mm)	28.93	46.46	100.22	126.21	103.30
	Temperature (°C)	11.20	12.20	12.20	12.00	11.80
Pasacunta-Qollpacanta						
Dry observed (1999)	Precipitation (mm)	2.50	5.50	77.80	115.50	74.50
	Temperature (°C)	8.20	9.70	11.20	11.10	11.30
Wet observed (2013)	Precipitation (mm)	64.00	25.00	78.50	79.90	43.10
	Temperature (°C)	8.70	10.30	10.30	10.70	9.60
Dry climate change	Precipitation (mm)	15.36	22.32	74.40	112.56	71.20
	Temperature (°C)	12.20	14.00	14.70	14.20	14.20
Wet climate change	Precipitation (mm)	24.96	36.27	120.90	182.91	115.70
	Temperature (°C)	10.70	12.50	13.20	12.70	12.70
Peñas Kerani						
Dry observed (1999)	Precipitation (mm)	20.00	32.00	51.90	92.60	73.00
	Temperature (°C)	8.80	8.70	8.10	7.40	9.80
Wet observed (2013)	Precipitation (mm)	47.80	60.30	129.10	105.80	68.00
	Temperature (°C)	9.30	10.40	10.90	10.20	10.40
Dry climate change	Precipitation (mm)	23.92	42.08	111.68	89.04	84.64
	Temperature (°C)	12.80	13.70	13.80	13.40	13.20
Wet climate change	Precipitation (mm)	38.87	68.38	181.48	144.69	137.54
	Temperature (°C)	11.30	12.20	12.30	11.90	11.70



Fig. 2. Schematic description of the elicitation procedure.

presented together with an irrigation technique scenario that varies by community (see Section 3.1.4).

In the third step, experts report the average potato yields they expect under the irrigation and agronomical scenarios described in step 1 and step 2.

In step 4, a climate change scenario is presented to the experts. This scenario is described in terms of precipitation and temperature conditions expected to hold during the cropping season. All experts are presented to four climate change scenarios for each community under study (see Section 3.1.5). These scenarios are unlabeled –i.e. the expert does not receive information on whether the weather conditions have been observed or are hypothetical.

In step 5, experts report their expectations about the potato yields attainable under the climate change scenario presented in step 4. Experts are asked to keep in mind the irrigation scenario presented in step 2, and the agronomical scenario described in step 1.

Experts answer in a dichotomous manner to the elicitation questions in step 5. That is, experts answer either yes or no to the following question:

1. Under this scenario, would you expect at least **initial %** of the average yields [reported by the expert in step 3]?

If the answer to question (1) is no, then a follow-up question decreases the percentage presented to the expert:

2a. Would you expect at least **if not follow-up** % of the average yields [reported by the expert in step 3]?

If the answer to question (1) is yes, a follow-up question increases the percentage of presented to the expert:

2b. Would you expect yields to surpass average yields [reported by the expert in step 3] by at least (**if yes follow-up %-100)%**?

Notice that the elicitation questions (1), (2a) and (2b) use the average value reported in step 3 as a pivot value. That is, the expert judges whether yields feasibly will be lower or higher than yields he/she reported under average precipitation and temperature conditions –and answers with a yes or a no.

Table 5 reports initial and follow up percentages –i.e. *initial %, if no follow-up %*, and *if no follow-up %* in questions (1), (2a), and (2b).

These percentages vary by irrigation and precipitation-temperature scenarios. These percentages do not vary across experts or communities under study but this is not a deterrent of the variation in the yields implied by the experts' dichotomous answers. First, the implied yields under climate change vary across experts regardless all experts answering to the same percentage changes because each expert uses his/her own average potato yields as reference. Also, the implied yields vary across communities because the reference average yields vary across communities.

The percentages reported in Table 5 have been designed to allow for the possibility that climate change either increase or decrease potato yields –i.e. either climate change may be beneficial or detrimental for a specific community. For instance, for the DO scenario in the first row of Table 5, the *if yes follow-up* percentage takes value 110% which implies that we are requesting from the expert to considerer whether dry conditions may increase potato yields by at least 10% with respect to the average yields. While this increase is unlikely under dry conditions, it may happen under wet conditions and that is why, for instance, the *initial* percentage for the WCC scenario is 110% (last row in Table 5). That is, we are asking from the expert to initially consider the plausibility that wet conditions may increase potato yields by at least 10%. If the answer is no, then we evaluate the possibility of a decrease of 30%. If the answer is yes, then we explore whether the expert may expect an increase of 40% in yields.

Table 5

Percentages presented to experts in elicitation question^a.

Scenario	No irrigation		Irrigatior	1		
	Initial ^b	Follow up		Initial ^b	Follow	up
		If no ^c	If yes ^d		If no ^c	If yes ^d
Dry observed Wet observed Dry climate change Wet climate change	80% 90% 0% 110%	40% 30% 50% 70%	110% 115% 130% 140%	90% 0% 105% 130%	70% 60% 70% 80%	125% 125% 140% 160%

^a Step 3, substep 3 in Fig. 2.

^b Initial question: Would you expect at least *initial* % of the average yields?

^c If no: Would you expect at least *if no follow-up* % of the average yields?

^d If yes: Would you expect yields surpass average yields by at least (if yes follow-up % - 100)%?

Also, by varying percentages across irrigation scenarios, we investigate whether yields are expected to be higher under irrigation and under climate change conditions. For instance, for the DO under no irrigation in the first row of Table 5, if the expert answers yes to the *initial* 80%, then the *if yes follow-up* question investigates whether a 10% increase in average yields is plausible. These percentages are higher for the irrigation scenario and the same DO scenario –90% and 25%, respectively— to allow for the possibility that irrigation is a mitigation tool.

Readers familiar with the non-market valuation literature may have noticed that our elicitation question heavily borrows from the doublebounded dichotomous question used in contingent valuation studies.² EE protocols usually request answers in the form of subjective probabilistic distributions (Morgan, 2014; Bosetti et al., 2016). However, our experts reported difficulty expressing their opinions directly in probability terms – which is not uncommon (see James et al., 2010). Thus, we decided to borrow the double-bounded dichotomous question – which proved to be more intuitive for our experts (see Section 5.1).

3.2. Statistical Approach to Aggregate Experts' Opinions

To summarize the opinions of our experts, we also borrow the maximum likelihood (ML) approach used in the contingent valuation literature to analyze data collected by a double-bounded dichotomous question (see Hanemann et al., 1991). Given a parametric distribution, the ML approach seeks the parameter values that maximize the probability of observing the gathered data.

To define the likelihood function of interest in this study, let W_{ejk} be the yields expected by expert e at community j under the average precipitation/temperature scenario and irrigation scenario k –step 3 in Fig. 2. Let Z^{bkc} be the percentage change presented by the elicitation questions –taking values reported in Table 5. The superscript b = i, n, ydenotes percentages presented to the expert in the elicitation questions: i refers to the *initial* percentage presented in question (1); n refers the percentage presented in question (2a) –the *if no follow-up* question—; and y refers to the percentage in question (2b) –the *if yes follow-up* question. The superscript k = r, t denotes the percentages presented under each irrigation scenario: r refers to no irrigation, and t refers to the irrigation technique scenario. The superscript c = do, wo, dcc, wcc denotes the climate change scenarios.

Then $W_{ejk}^{bc} = W_{ejk}(Z^{bkc}/100)$ are the yields implicit in the percentage changes presented by the elicitation questions. While the percentage changes Z^{bkc} are constant across communities and experts, the implicit yields W_{ejk}^{bc} vary across experts and communities because each expert's average yields (W_{ejk}) are used in the calculation of W_{ejk}^{bc} .

Finally, let \overline{W}_{ejk}^c be the yields that expert *e* judges feasible to be observed at community *j* under irrigation scenario *k* and climate change scenario *c*. Then the likelihood function of our interest is defined as follows

$$\begin{split} \ln\left(\mathcal{L}\right) &= \sum_{e} \sum_{j} \sum_{k} \sum_{c} \left\{ d_{j}^{NN} \ln\left[Pr\left(\overline{W}_{ejk}^{c} < W_{ejk}^{nc}\right) \right] \\ &+ d_{j}^{YN} \ln\left[Pr\left(W_{ejk}^{c} < \overline{W}_{ejk}^{c} < W_{ejk}^{yc}\right) \right] \\ &+ d_{j}^{NY} \ln\left[Pr\left(W_{ejk}^{nc} < \overline{W}_{ejk}^{c} < W_{ejk}^{ic}\right) \right] \\ &+ d_{j}^{YY} \ln\left[Pr\left(\overline{W}_{ejk}^{c} < W_{ejk}^{yc}\right) \right] \right\}, \end{split}$$

where \mathcal{L} denotes the log of the likelihood function; d_j^{NN} takes value one if the expert has answered no to the initial elicitation question (1) and no to the if no follow-up question (2a), and zero otherwise; d_j^{YN} , d_j^{NY} , and d_j^{YY} are similarly defined for the sequences yes-no, no-yes and yes-yes. Pr(.) refers to the probability that the event defined in parenthesis is considered a feasible event by the expert. For instance, $\Pr(W_{ejk}^{ic} < \overline{W}_{ejk}^{c} < W_{ejk}^{yc})$ refers to the probability that the yields judged feasible by expert *e* for community *j* under scenarios *k* and *c* (i.e. \overline{W}_{ejk}^{c}) fall in the range of values defined by the yields implicit in the initial question (1) and the if yes follow-up question (2b) –i.e. $[W_{ejk}^{ic}, W_{ejk}^{yc}]$.

The probabilities Pr(.) are modeled as arising from an extreme value distribution –which is the underlying distribution of a logit model. Because yields cannot be smaller than zero, we adjust the distribution to be zero-truncated. Also, we truncate the distribution from above because, arguably, there is a limit to the yields that can be observed in the local social-ecological systems in the Bolivian Altiplano. The maximum yields have been set at 20 –which is close to the maximum reported by experts under a wet climate change scenario (19.5 tons/ha). Details about the functional form of a truncated extreme value distribution are provided by Xu et al. (1994) who model, in a similar fashion than us, ranch land prices.

4. Summary of Experts' Opinions

The statistical approach described in Section 3.2 allows us to infer potato yields, conditioning on the factors described in the elicitation scenarios. We want to underline the following point: the conditional yields reported in this section correspond to those that experts judge feasible under the scenarios presented in the elicitation protocol. More specifically, the results in this section should not be interpreted as characterizing or modeling or defining the state of the potato yields in the Bolivian Altiplano.³ Instead, the yields reported in this section reflect a summary of the opinions that experts have provided when requested to consider the specific conditions described in the elicitation scenarios.

4.1. Conditional Yields

Table 6 reports the conditional yields implicit in the experts' dichotomous answers. We present two sets of conditional yields. The first set is obtained from an Ordinal Least Squares (OLS) specification. The second set is obtained from maximizing the likelihood function described in Section 3.2. Presenting results from an OLS specification is a common strategy in the contingent valuation literature to check for consistency of the results delivered by the maximization of a likelihood function – e.g. Madani et al. (2013) and Cooper and Signorello (2008).

Both statistical models are informed with 248 observations, gathered from 12 experts –8 experts provided 24 answers; 3 experts provided 16 answers; and 1 expert provided 8 answers.

The dependent variable in the OLS specification corresponds to the yields implicit in the follow-up elicitation questions. Specifically, yields are assigned as follows: i) zero if the expert answered the sequence no-no to the elicitation questions; ii) the yields implicit in question (1) if the expert answered yes-no; iii) the yields implicit in question (2a) if the expert answered no-yes; and iv) the yields implicit in question (2b) if the expert answered yes-yes.

In both specifications, yields are conditioned on i) the community under analysis –i.e. we obtain average yields for each community–; ii) irrigation scenarios –via dichotomous variables defining whether the expert has answered to a no irrigation scenario or to an irrigation scenario; iii) climate change scenarios –via dichotomous variables identifying whether the scenario is dry observed (DO), wet observed (WO), dry climate change (DCC), or wet climate change (WCC)–; iv) two sets of interaction variables resulting from interacting the climate change scenarios and the community identifiers, and the irrigation scenarios and the community identifiers; and v) an heuristic factor –via a dichotomous variable identifying whether the expert is affiliated to a NGO.

² A double-bounded dichotomous question seeks to establish the range in which an individual's willingness to pay (WTP) falls. It does so by sequentially evaluating whether the WTP is larger or smaller than two values randomly assigned to respondents (Hanemann et al., 1991). The second value is chosen by the enumerator conditional on the answer to the first value –i.e. if the first response is "yes", the second value is larger; if the first response is "no", the second value is smaller.

³ We thank the anonymous reviewer who pointed the need to clarify this point.

Table 6

Conditioning factors	Ordinal Least Squares (OLS) ^a		Maximum likelihood ^b	
	Conditional yields	Standard errors	Conditional yields	Standard errors
Communities ^c				
Intercept	2.56	(0.63)***	3.24	(0.44)***
Suriri-Capiri (SC)	2.01	(1.58)	1.14	(0.63)*
Pasacunta-Qollpacanta (PQ)	-0.75	(1.24)	-1.89	(0.65)***
Irrigation scenarios ^d				
Irrigation technology	3.25	(0.58)***	3.55	(0.31)***
Precipitation/temperature				
scenarios ^e				
Wet climate change (WCC)	2.15	(1.14)**	2.12	(0.67)***
Dry climate Change (DCC)	0.34	(0.71)	0.43	(0.63)
Wet observed (WO)	-0.61	(0.86)	-0.57	(0.66)
Interactions				
WCC * SC	-1.15	(1.31)	-1.24	(0.95)**
WCC * PQ	0.52	(1.02)	1.78	(0.97)**
DCC * SC	-1.89	(0.96)*	-2.30	(0.91)***
DCC * PQ	-0.51	(1.02)	0.61	(0.95)
WO * SC	-2.42	(1.63)	-2.36	(1.01)***
WO * PQ	0.87	(1.44)	2.94	(1.01)***
Irrigation * SC	-0.99	(0.99)	-0.23	(0.63)
Irrigation * PQ	1.58	(1.05)	0.97	(0.54)*
Heuristics				
Expert is affiliated to a NGO	5.66	(1.06)***	6.72	$(0.44)^{***}$
R-squared	0.401		-	
Log-likelihood	-		-488	
Number of observations ^f	248			
Number of experts ^f	12			

Standard errors are clustered by expert.

Significant at *90% confidence level, **95% confidence level, and ***99% confidence level. ^a For the OLS specification, the dependent variable is potato yields assigned as follows: zero if the expert answered no-no; the yields implicit in question (1) if expert answered yes-no; the yields implicit in question (2a) if expert answered no-yes; and the yields implicit in question (2b) if expert answered yes-yes.

^b Likelihood function, as explained in Section 3.2, is truncated from below at zero and from above at 20.

^c Reference category: Peñas Kerani scenario.

^d Reference category: no irrigation scenario.

^e Reference category: dry observed scenario.

^f 8 experts provided 24 answers; 3 experts provided 16 answers; and 1 expert provided 8 answers, for a total of 248 observations.

The first set of conditional yields in Table 6 is estimated using the OLS specification. The first row reports the intercept parameter which reflects the average yields in Peñas Kerani -i.e. 2.56 tons/ha. The second and third rows report the yields specific to the other two communities but the lack of statistical significance implies that, according to the OLS specification, yields do not significantly differ across communities. The next significant parameter in Table 6 captures the effect from the presence of an irrigation technique -an average increase of 3.25 tons/ha across communities. In terms of the potential impacts from climate change scenarios, experts expect that the WCC conditions increase yields by 2.15 tons/ha; and only Suriri-Capiri is expected to be impacted negatively by the DCC conditions -with a decrease of 1.89 tons/ha. The statistical significance of the variable identifying the NGO experts is interpreted as evidence of heuristic bias -i.e. NGO experts tend to report yields that are 5.66 tons/ha higher than the yields reported by other experts.

The second set of conditional yields in Table 6 result from maximizing the likelihood function in Section 3.2. The coefficients that are statistically significant in the OLS specification remain significant in the maximum likelihood (ML) approach, with identical signs but different point values. That is, average yields in Peñas Kerani are 3.24 instead of 2.56; the average increase due to irrigation is 3.55 instead of 3.25; the average increase due to WCC conditions is 2.12 instead of 2.15; the impact from the DCC on Suriri-Capiri is -2.30 instead of -1.89; and the heuristic bias is 6.72 instead of 5.66. This similarity in results is interpreted as evidence of consistency of the estimates. Several factors that are not statistically significant in the OLS specification become significant in the ML approach –due mostly to a decrease in the standard errors which is an advantage from using the double-dichotomous methodology (Hanemann et al., 1991).

Thus, the ML approach allows us to gain further insights into the heterogeneity in the expected effects from climate change. For instance, in contrast to the OLS, the ML approach delivers differences in the average yields across communities. Importantly, we learn that experts consider that average yields are the lowest in Pasacunta-Qollpacanta – 1.89 tons/ha less than the 3.24 tons/ha in Peñas Kerani. Also, Suriri-Capiri is the community with the highest average yields in the opinion of the experts –1.14 tons/ha more than the 3.24 tons/ha in Peñas Kerani.

4.2. Impacts From Climate Change

The ML approach delivers insights into the experts' opinions about the direction and heterogeneity of climate change impacts at the local scale. Consistently with the OLS analysis, the ML approach documents that experts expect positive impacts from the WCC conditions. In addition, the ML approach delivers differences in the impacts from WCC across communities. For instance, while the effect from WCC conditions is expected to be 2.12 tons/ha in Peñas Kerani, this same positive effect is expected to be of only (2.12 - 1.24 =) 0.88 tons/ha in Suriri-Capiri. In contrast, WCC conditions are expected to increase yields by 3.90 tons/ha in Pasacunta-Qollpacanta. Thus, the positive impact from wet hypothetical conditions is expected to be largest at Pasacunta-Qollpacanta and smallest at Suriri-Capiri.

With respect to the effects from WO conditions, experts expect a negative effect in Suriri-Capiri (-2.36), a positive effect in Pasacunta-Qollpacanta (2.94), and no effect in Peñas Kerani. Thus, according to experts' opinions, wet observed conditions have impacted positively Pasacunta-Qollpacanta but negatively Suriri-Capiri.

Notice the consistency of the raking implied under both wet scenarios: Pasacunta-Qollpacanta is expected to benefit from wet climate change conditions; Peñas Kerani is expected to benefit moderately or not at all from wet conditions; and Suriri-Capiri may benefit but also may be damaged by wet climate change conditions.

With respect to the effects from DCC conditions, experts expect negative impacts on yields only in Suriri-Capiri (-2.30 tons/ha). Thus, Suriri-Capiri is not only expected to be damaged by wet climate change conditions buy also by dry ones.

4.3. Are Positive Effects From Irrigation Enough to Mitigate the Negative Effects From Climate Change?

The OLS and the ML approach deliver similar average positive impacts from irrigation across communities –an increase of 3.25 tons/ha (OLS) or 3.55 tons/ha (ML). And only the ML approach deliver a community-specific impact from irrigation –an increase of 0.97 tons/ha (additional to the average impact of 3.55 tons/ha) in Pasacunta-Qollpacanta.

We explore whether these positive effects from irrigation are enough to mitigate the negative impacts from climate change –i.e. whether experts expect that a community with irrigation and under climate change conditions will keep average yields at no irrigation, no climate change average levels.

Thus, we calculate potato yields under the four climate change scenarios and assume that an irrigation technique is in place. We also calculate the 95% confidence intervals because we want to check whether these conditional yields are different from average yields in a baseline scenario.⁴ We choose the no irrigation, no climate change

⁴ Conditional yields and their corresponding empirical 95% confidence intervals are estimated by taking 5000 Krinsky-Robb draws from a multivariate normal distribution. The vector of means and the covariance matrix of this multivariate distribution correspond to the coefficients reported in Table 6 for the ML approach and their corresponding covariance matrix (see Krinsky and Robb (1986) for details).



Fig. 3. Average yields (tons/ha) in Suriri-Capiri and 95% confidence interval by irrigation and climate change scenarios.

situation as baseline. Yields under this baseline scenario are inferred from experts' answers to no irrigation conditions and the average weather conditions presented in steps 1 and 2 of the elicitation procedure (see Fig. 2).

Importantly, when estimating these conditional yields, we assume that the dichotomous variable identifying the NGO experts takes value zero –i.e. we do not include in our estimations the extra yields that ONG experts tend to expect. This decision is discussed in Section 5.4.

Figs. 3 to 5 depict average yields and 95% confidence intervals. The grey symbols refer to the average yields (square), and lower and upper bounds (horizontal bars) under no irrigation scenarios. The black symbols refer to the average yields (circle), and lower and upper bounds (horizontal bars) under irrigation scenarios. From left to right, the first set of results refers to the 1990–2013 weather conditions presented as average to the experts. The other four sets of results refer, respectively, to the DO, WO, DCC and WCC scenarios.

Fig. 3 depicts results for Suriri-Capiri –which, according to the analysis in Section 4.2, may benefit or may be damaged by climate change. The baseline average yields in Suriri-Capiri are 5.5 tons/ha, with a 95% confidence interval ranging from 3.76 to 7.21. For the DO conditions, Fig. 3 depicts a situation in which stoned-lines channels may not be the most effective irrigation technique to significantly boost yields. On one hand, DO conditions are not expected to dramatically impact yields —if Suriri-Capiri faces the DO conditions, experts judge that yields will fall within the 3.86 and 5.55 interval which is a statistically insignificant reduction with respect to the baseline yields. On the other hand, if stoned-lines channels are present under DO conditions, the extra yields are not enough to surpass baseline levels –experts judge that yields will fall within the 7.07 and 8.88 interval which mostly fall above the upper bound of the baseline yields but is still not enough to produce a statistical difference.

For the WO conditions, Fig. 3 depicts a situation in which stonedlines channels represent an effective measure to mitigate the impacts from wet climate change conditions. On one hand, WO conditions are expected to drive yields below baseline levels if no irrigation is in place –falling between 0.73 and 2.78 which represents a reduction in yields with respect to the baseline values (3.76 to 7.21). On the other hand, if stoned-lines channels are present under WO conditions, the extra yields expected to fall between within 4.08 and 5.98 –range that falls within the 95% confidence interval of the baseline yields. That is,



Fig. 4. Average yields (tons/ha) in Pasacunta-Qollpacanta and 95% confidence interval by irrigation and climate change scenarios.

- Upper bound	d, No irrigation — Lo	ower bound, no irrigation	Mean, No irrigation	
- Upper bound	Jpper bound, Irrigation Lower bound, irrigation		 Mean, Irrigation 	— 10.04
9.26				• 9.06
			8.23	8.09
• 7.4	— 7.51	7.38	• 7.33	
0.9	• 6.51	• 6.43	6 .4	6.56
5.54	5.52	5.52		5.56
5.4			1005 4.7	4.56
3.73	4.01	inna 3.85	3.83	
	3.01	2.93	2.96	
	2.02	2.01		
1990-2013	Dry Observed	Wet Observed	Dry CC	Wet CC

Fig. 5. Average yields (tons/ha) in Peñas Kerani and 95% confidence interval by irrigation and climate change scenarios.

yields under WO conditions are expected to fall below baseline yields, and stoned-lines channels are expected to return yields to values similar to those in the baseline scenario.

Following a similar reasoning for the DCC, Fig. 3 also illustrates that stoned-lines channels represent an effective mitigation tool under DCC conditions. For the case of WCC conditions, while the stoned-lines channels are effective at boosting yields, the WCC conditions are not expected to impact yields. This situation opens space to argue that, if no impact from climate change is expected under WCC, then irrigation techniques are not needed to mitigate climate change –argument that does not preclude the improvement of irrigation techniques for productivity purposes.

Fig. 4 depicts results for Pasacunta-Qollpacanta – which, according to the analysis in Section 4.2, is expected to be benefited by wet climate change conditions. The baseline yields are 4.43 tons/ha with a 95% confidence interval ranging from 3.07 to 5.57.

Fig. 4 illustrates that stone-lines channels in Pasacunta-Qollpacanta are expected to be an effective mitigation tool under dry scenarios (DO and DCC) –yields are expected to fall below the 95% baseline confidence interval but irrigation returns yields to baseline levels. For the wet scenarios, Fig. 4 illustrates that yields may not be negatively impacted by wet scenarios and, consequently, irrigation techniques may not be needed for mitigation purposes Again, this does not exclude the possibility that irrigation may be a good strategy to boost productivity – which seems to be the case under both wet scenarios, as illustrated by Fig. 4.

Finally, Fig. 5 depicts results for Peñas Kerani –which, according to the analysis in Section 4.2, is expected to be moderately impacted by climate change conditions. The baseline yields are 5.40 tons/ha with a 95% confidence interval ranging from 3.73 to 6.90.

Fig. 5 illustrates that open channel irrigation is expected to be an effective mitigation tool in Peñas Kerani. Three climate change conditions (DOO, WOO, and DCC) are expected to have negative effects, and WCC is expected not have no effects on yields. Under the three scenarios provoking negative impacts, open channel irrigation is expected to increase yields to baseline values.

5. Methodological Discussion

In this section, we discuss methodological issues that are arise in expert elicitation studies. This discussion is particularly important to this study because the format of our elicitation question doses not follow the conventional strategy of asking experts to report their opinions in distributional terms.

5.1. Motivation for Our Elicitation Question

We first discuss the motivation to borrow the double-dichotomous choice elicitation question. One main challenge of expert elicitation is its reliance on individuals that are experts on a given field but are not necessarily proficient at expressing their opinions in distributional terms (Bosetti et al., 2016). Conventionally, experts report their opinions in terms of probability distributions (e.g. Al-Awadhi and Garthwaite, 2006; Denham and Mengersen, 2007; Gill and Walker, 2005). However, even experts that are familiar with the concept of probability may find difficult to express their opinions in probability terms (James et al., 2010).

Experts consulted in this study are not the exception. Our experts, although proficient in probability and statistics, did not feel comfortable expressing their opinions in the form of probability distributions. We first realized this when designing the pilot elicitation questions with the help of Agronomist Corina Apaza –who was hired to assist the research team in, among other tasks, the design of the elicitation questions. Then, a couple of conventional versions of the EE questions were piloted and explained to potential experts during the personal visits paid to them in January 2014 –six months in advance to the implementation of the protocol. It became clear that reporting opinions in distributional terms was not a straightforward task for our experts.

Recognizing the similarities between the contingent valuation studies and the expert elicitation applications, we decided to test the feasibility of borrowing the double-bounded dichotomous question from the contingent valuation literature (see Hanemann et al., 1991). Initially, we tested the possibility of evaluating the changes in yields directly – that is, presenting changes in tons/ha units. However, this strategy did not prove straightforward neither. Then, we arrived to the format that we use in this study: changes in yields using as pivot the experts' own expectation of yields in the baseline scenario.⁵ Our research assistant and one expert answering a pilot reportedly found this elicitation format intuitive and easier.

Why is that experts find easier to answer in a dichotomous way to changes presented in percentage terms? We believe that there is a

⁵ Readers familiar with design of discrete choice experiments may notice the similarity of our strategy and the pivot design implemented in the Stated Choice literature (see Hess and Rose, 2009).

simple reason: in their professional careers, our experts frequently communicate judgements about percentage changes in a dichotomous manner. For instance, given a local social-ecological system, our experts recurrently judge whether an extra kg/ha of fertilizer may increase crop yields by 10%. They may also be able to judge whether an extra kg/ha increases yields by 1 ton/ha. However, they prefer to use a reference point in their judgement –e.g. last season's yields. A reference point helps because, among other things, the relations we are talking about are not linear. Economists are not unfamiliar with this situation –e.g. an economist may feel more comfortable providing opinions about price elasticities than about unitary changes in prices.

5.2. Cognitive Heuristics and Bias

We have argued in Section 5.1 that our elicitation question is advantageous because it is intuitive to our experts –resembling decision situations that they recurrently face. However, this elicitation format has implications in terms of anchoring.

It has been documented that, when people are presented to a starting value that they are asked to adjust, they typically do not adjust sufficiently –a cognitive heuristic known as anchoring (Morgan, 2014). If we were eliciting probability distributions, a standard procedure to minimize the influence of this heuristic is to ask first for the tails of the distributions.

But we are not directly gathering probability distributions. Instead, we are requesting that the experts adjust yields taking as reference value their own expectations on yields under average conditions. Thus, we may be inducing experts to anchor their expectations at their baseline yields.

We argue that anchoring is not a troublesome feature in this application. This is so because we are not interested on values falling in the tails of the yields' distribution. While we study the impacts of climate change, we have designed climate change scenarios that do not belong to the tails of the distribution of the climatic events –two scenarios have actually occurred in recent years. Consequently, impacts on the yields are not expected to fall in the tails of the distribution of yields.⁶

An additional cognitive heuristic, known as availability, implies that people assess the probability of an event by the ease with which examples can be brought to mind (Morgan, 2014). The usual strategy to minimize the chance of availability impacting experts' opinions is to encourage experts to systematically consider all relevant evidence. During the implementation of our protocol, experts were encouraged to exchange points of view during the first section of the workshop –during which experts were provided with the goal of the elicitation and background information. Then, experts were encouraged to answer the protocol taking into consideration what had been discussed but without consulting to each other during the last 2 h of the workshop.

5.3. Overconfidence

Overconfidence is pervasive in expert elicitation studies (Morgan, 2014). Overconfidence implies that experts state probability intervals that are poorly calibrated in the sense that they are too narrow –i.e. the realized value falls outside the central intervals much more frequently than they should (Bosetti et al., 2016).

We argue that the format of our elicitation question takes the burden of overconfidence away from the expert and put it on the researcher. This is so because the maximum length of the tails of the distributions are implicitly defined when deciding the percentage changes presented to the experts. As listed in Table 5, we aim to cover long tails by allowing not only for damages but also benefits from climate change. So, for instance, we asked experts if they would expect an increase of 60% in the yields under wet climate change scenarios.

The fact that the researcher implicitly decides the length of the tails of the distribution can be seen as a drawback. However, as with the anchoring issue, we argue that this application is less interested on the tails of the distributions and more on the central values.

5.4. Aggregation and Weighting of Experts' Opinions

We now discuss aggregation of expert's opinions –an issue that concerns to the analysis and presentation of the data gathered via an expert elicitation (Bosetti et al., 2016). Aggregation is attained either through mathematical strategies, or through a behavioral approach in which experts are requested to reach an agreement regarding the final values (Bolger and Rowe, 2015; Riabacke et al., 2012). The main motivation to aggregate experts' opinions is to inform policy makers (Bolger and Rowe, 2015).

Regardless the type of aggregation, a controversial issue is whether judgements of all experts should be weighted equally or not (Bolger and Rowe, 2015). Partially due to the controversy of weighting, some practitioners have proposed that documenting the heterogeneity in the experts' opinions is more informative than aggregated numbers (Morgan, 2014; Winkler, 2015). They argue that, because aggregation may hide the richness in opinions, heterogeneity in itself is relevant information for policy makers.

In this application we do aggregate but also document, to some extent, the heterogeneity in opinions –both across experts and across communities under study. Indeed, our aggregation approach is a mathematical one –explained in Section 3.2. We are able to document heterogeneity across experts because we identify that NGO experts tend to estimate higher yields than the other experts. We actually exclude these extra yields when estimating the values expected under climate change scenarios. By doing so, we are underweighting NGO experts. We do so because the aggregated numbers that we report are, arguably, useful to policy makers and we still can inform them about the heterogeneity in the opinions.

5.5. Is This an Expert Elicitation or a Contingent Valuation Applied to Experts?

We believe it is worth spending a few lines in the discussion of whether our application is an expert elicitation that borrows and adapts a question from the contingent valuation, or a contingent valuation study applied to experts.

We argue that the objective nature of the elicited quantity is the feature that makes this study an expert elicitation. That is, this study gathers subjective opinions from experts about the realization of an objective quantity (yields). A contingent valuation study, in contrast, gathers the subjective evaluation of an individual about a subjective quantity (willingness to pay).

A second reason to deem this study as an expert elicitation application is that experts' are familiar with the type of judgements we are requesting from them. They may not be used to communicate them through a protocol but they do recurrently carry out judgements about yields and their association with weather conditions. In contrast, respondents of a contingent valuation study are put under a scenario that is foreign to them –because contingent valuation is usually a last resource to simulate a non-existent market.⁷

⁶ There are examples of previous applications that trade anchoring for simplicity. For instance, Landeta (2006) gathers the judgement of experts about tourist expenditures in Catalonia. This application, like ours, is mostly interested on the distribution around the average values than on the tails of the distribution. The tails would provide information about expenditures of outliers. But, in the context analyzed by Landeta (2006), information about the tails is less useful than a robust estimation of average tourist expenditures.

⁷ We would like to direct the attention of the reader to a nascent literature that combines expert elicitation and non-market valuation: Leon et al. (2003), Alberinni et al. (2006), Van Houtven et al. (2014), Ahtiainen and Martinez-Cruz (2016), and Strand et al. (2017). Our study somehow relates to this literature, in the sense that intersections between expert elicitation and non-market valuation are explored. However, the topics of analysis of these papers, with the exception of Alberinni et al. (2006), do not directly relate to ours.

6. Conclusions

This paper explores whether expert elicitation is a useful tool to gain insights about impacts from climate change on crop yields in local social-ecological systems –for which good quality information is seldom available. In addition, we explore whether our data provide insights on the effectiveness of irrigation in mitigating the negative impacts from climate change.

In particular, we have gathered the opinions of experts about the impact from climate change on potato yields in three communities located in the Bolivian Altiplano. These communities are Suriri-Capiri, Pasacunta-Qollpacanta, and Peñas Kerani. We request from experts their judgements about the yields that would hold under four precipitation/temperature scenarios. These scenarios have been designed to resemble climate change conditions discussed in two documents: an *official* Bolivian document, and a report from the World Bank on the impacts from climate change in Bolivia. While these scenarios are not labeled when presented to the experts, we label them here as wet and dry –two of each type.

Our data allow us to document local heterogeneity in the climate change impacts expected by experts. In average, wet climate change scenarios are expected to positively impact yields, and dry climate change scenarios are expected to decrease yields. However, these average effects hide substantial heterogeneity across the communities under study. For instance, Pasacunta-Qollpacanta is expected to benefit from wet climate change conditions; Suriri-Capiri may benefit or may be damaged by wet climate change scenarios; and Peñas Kerani is expected to not be impacted by wet conditions. In contrast, only Suriri-Capiri is expected to experience losses under dry climate chance scenarios, and the other two communities are expected to experience no changes at all.

Our data also allow us to document heterogeneity in the expected effectiveness of irrigation as a mitigation tool. Effectiveness varies not only across communities but also across climate change scenarios. For instance, under dry scenarios, irrigation is clearly effective in Pasacunta Qollpacanta but not so in Suriri-Capiri.

This study is ultimately motivated by the need to inform public policies with the best available scientific knowledge. Indeed, our study delivers helpful insights for policy makers in Bolivia. On one hand, our results suggest that irrigation is an effective tool to mitigate the effects from dry climate change scenarios. On the other hand, because wet scenarios may benefit at least one of these communities, policy makers may want to reconsider the encouragement of irrigation for mitigation purposes in this community –which does not exclude the possibility that policy makers finance irrigation projects for productivity reasons. Clearly identifying the purpose of a policy is essential to implement evidence-based policies –which in this case implies that policy makers are informed that irrigation may not be needed for mitigation purposes but for productivity ones.⁸

A caveat is in place. Given our data gathering strategy, the results of this paper cannot be interpreted as characterizing or defining the state of the potato yields in the Bolivian Altiplano. Our results aggregate the most informed judgements about how yields react to the specific conditions described in the elicitation scenarios. The elicitation scenarios capture realistic features but, by design, remain simple and manageable. Thus, as with any other model that simplifies reality, our results hold within the limits of the scenarios defined in our elicitation procedure, and generalizations are not recommendable.

Despite this caveat, we believe that our data collection strategy represents a promising decision support tool for a wide range of public policy issues. Particularly, when monetary and time constraints converge with the lack of scientific information. For instance, State authorities in Mexico usually have less than three months to select the municipalities that should be covered by a government-sponsored index insurance. Indeed, these decisions have been taken in the absence of a strategy to gather local data (see Martinez-Cruz et al., 2016). A second example refers to the gathering of information about overuse of watersheds. In this context, authorities may be interested in learning which strategies are expected to contribute the most to preserve a watershed. A straightforward way to adapt this paper's methodology is as follows: in the first round of questions, ask the experts for their judgement of the gap between supply and demand of water in the watershed –this question is pertinent because there is usually uncertainty about the magnitude of this gap. Then the elicitation questions would refer to percentage changes in the gap due associated with specific water conservation strategies.⁹

Our data collection strategy represents an alternative to incorporate the opinions from experts into methodologies that also gather local information –e.g. participatory strategies (Bentley Brymer et al., 2016) and citizen science methods (Bonney et al., 2009). For instance, Van Aalst et al. (2008) explore the use of community risk assessment in the context of climate change adaptation. Our methodology could complement a participatory study with similar goals than Van Aalst et al.'s (2008) by suggesting risk management instruments that have been pre-selected based on experts' opinions and need to be adapted to local governances.

References

- Ahtiainen, H., Martinez-Cruz, A.L., 2016. Using expert elicitation to calibrate benefit transfers: an application to the Baltic Sea. Presented at the 22nd Annual Conference of the European Association of Environmental and Resource Economists, Zurich, Switzerland, 22–25 June Available at. http://www.webmeets.com/files/papers/EAERE/ 2016/1267/BT_EE_EAERE16_new.pdf.
- Al-Awadhi, S.A., Garthwaite, P.H., 2006. Quantifying expert opinion for modelling fauna habitat distributions. Comput. Stat. 21 (1), 121–140.
- Alberinni, A., Chiabai, A., Muehlenbachs, L., 2006. Using expert judgment to asses adaptive capacity to climate change: evidence from a conjoint choice survey. Glob. Environ. Chang. 16 (2), 123–144.
- Andersen, L., Mamani, R., 2009. Cambio climático en Bolivia hasta 2100: Síntesis de costos y oportunidades. Estudio Regional de Economía del Cambio Climático en Sudamérica http://inesad.edu.bo/bcde2009/A2%20Andersen%20Mamani.pdf (Accessed 25 Nov 2016).
- Baker, E., Bosetti, V., Jenni, K.E., Ricci, E.C., 2014. Facing the experts: survey mode and expert elicitation. FEEM Working Paper No. 1.2014 https://papers.csm?sol3/papers.cfm?abstract_id=2384487 (Accessed 30 Nov 2016).
- Bentley Brymer, A., Holbrook, J., Niemeyer, R., Suazo, A., Wulfhorst, J.D., Vierling, K., ... Rachlow, J., 2016. A social-ecological impact assessment for public lands manage-
- ment: application of a conceptual and methodological framework. Ecol. Soc. 21 (3). Bolger, F., Rowe, G., 2015. The aggregation of expert judgment: do good things come to those who weight? Risk Anal. 35 (1), 5–11.
- Bonney, R., Cooper, C.B., Dickinson, J., Kelling, S., Phillips, T., Rosenberg, K.V., Shirk, J., 2009. Citizen science: a developing tool for expanding science knowledge and scientific literacy. Bioscience 59 (11), 977–984.
- Bosetti, V., Anadón, L.D., Baker, E., Reis, L.A., Verdolini, E., 2016. The future of energy technologies: an overview of expert elicitations. GGKP Working Paper 01 http://www. greengrowthknowledge.org/sites/default/files/downloads/resource/The_Future_of_ Energy_Technologies_An_Overview_of_Expert_Elicitations_0.pdf (Accessed 30 Nov 2016).
- Cooke, R.M., Goossens, L.H.J., 2000. Procedures guide for structural expert judgement in accident consequence modelling. Radiat. Prot. Dosim. 90 (3), 303–309.
- Cooper, J.C., Signorello, G., 2008. Farmer premiums for the voluntary adoption of conservation plans. J. Environ. Plan. Manag. 51 (1), 1–14.
- Denham, R., Mengersen, K., 2007. Geographically assisted elicitation of expert opinion for regression models. Bayesian Anal. 2 (1), 99–135.
- Folke, C., Hahn, T., Olsson, P., Norberg, J., 2005. Adaptive governance of social-ecological systems. Annu. Rev. Environ. Resour. 30, 441–473.
- Gill, J., Walker, L.D., 2005. Elicited priors for Bayesian model specifications in political science research. J. Polit. 67 (3), 841–872.
- Hanemann, M., Loomis, J., Kanninen, B., 1991. Statistical efficiency of double-bounded dichotomous choice contingent valuation. Am. J. Agric. Econ. 73 (4), 1255–1263.
- Hess, S., Rose, J.M., 2009. Should reference alternatives in pivot design SC surveys be treated differently? Environ. Resour. Econ. 42 (3), 297–317.
- Inter-American Development Bank (IADB), 2013. Call for Research Proposals Assessing Local Vulnerability to Climate Change in Latin America and the Caribbean. http://

⁸ Indeed, this argumentation points to a related but different research topic: underlying this argument, we are assuming that policy makers can assign probabilities to the occurrence of a given climate change scenario. This is hardly the case and opens the possibility of eliciting probabilities of occurrences of local climate change scenarios.

⁹ Ongoing work on this issue is being carried out by Sainz-Santamaria and Martinez-Cruz (2017).

idbdocs.iadb.org/wsdocs/getdocument.aspx?docnum=38010768 (Accessed 30 Nov 2016).

- Inter-American Development Bank (IDB), 2008. Programa Nacional de Riego con Enfoque de Cuenca (BO-L1021). Perfil del proyecto. http://idbdocs.iadb.org/wsdocs/ getdocument.aspx?docnum=1386686 (Accessed 25 Nov 2016).
- James, A., Choy, S.L., Mengersen, K., 2010. Elicitator: an expert elicitation tool for regression in ecology. Environ. Model. Softw. 25 (1), 129–145.
- Krinsky, I., Robb, A.L., 1986. On approximating the statistical properties of elasticities. Rev. Econ. Stat. 68 (4), 715–719.
- Landeta, J., 2006. Current validity of the Delphi method in social sciences. Technol. Forecast. Soc. Chang. 73, 467–482.
- Leon, J.C., Vazquez-Polo, F., Leon-Gonzalez, R., 2003. Elicitation of expert opinion in benefit transfer of environmental goods. Environ. Resour. Econ. 26, 199–210.
- Lybbert, T.J., Sumner, D.A., 2012. Agricultural technologies for climate change in developing countries: policy options for innovation and technology diffusion. Food Policy 37 (1), 114–123.
- Madani, S., Martínez-Cruz, A.L., McConnell, K.E., 2013. Conservation value of coral reefs around Kish Island, Iran. Mar. Resour. Econ. 28 (4), 331–343.
- Martinez-Cruz, A.L., Leal-Cota, V., Perez-Sosa, L., 2016. Insuring Federal Government Budget Against Catastrophic Events: The Mexican Catastrophe Climate Contingency Insurance Program (Mimeo).
- Ministerio de Medio Ambiente y Agua de Bolivia (MMAAB), 2009. Segunda Comunicación Nacional del Estado Plurinacional de Bolivia ante la Convención Marco de las Naciones Unidas sobre Cambio Climático. Programa Nacional de Medio Ambiente http://unfccc.int/resource/docs/natc/bolnc2exsums.pdf (Accessed 25 Nov 2016).
- Mistry, J., Berardi, A., Tschirhart, C., Bignante, E., Haynes, L., Benjamin, R., ... Jafferally, D., 2016. Community owned solutions: identifying local best practices for social-ecological sustainability. Ecol. Soc. 21 (2):42. http://dx.doi.org/10.5751/ES-08496-210242.
- Morgan, M.G., 2014. Use (and abuse) of expert elicitation in support of decision making for public policy. Proc. Natl. Acad. Sci. 111 (20), 7176–7184.
- Ostrom, E., 2009. A general framework for analyzing sustainability of social-ecological systems. Science 325 (5939), 419–422.
- Riabacke, M., Danielson, M., Ekenberg, L., 2012. State-of-the-art prescriptive criteria weight elicitation. Adv. Decis. Sci. 2012. http://dx.doi.org/10.1155/2012/276584.
- Sainz-Santamaria, J.J., Martinez-Cruz, A.L., 2017. Would Spreading Sprinkler Irrigation be the Solution to Aquifer Overdraft?: Insights From an Expert Elicitation in Aguascalientes, Mexico (Mimeo).

- Shaw, A., Sheppard, S., Burch, S., Flanders, D., Wiek, A., Carmichael, J., ... Cohen, S., 2009. Making local futures tangible–synthesizing, downscaling, and visualizing climate change scenarios for participatory capacity building. Glob. Environ. Chang. 19 (4), 447–463.
- Strand, J., Carson, R.T., Navrud, S., Ortiz-Bobea, A., Vincent, J.R., 2017. Using the Delphi method to value protection of the Amazon rainforest. Ecol. Econ. 131, 475–484.
- United Nations Framework Convention on Climate Change (UNFCCC), 2015. Adoption of the Paris Agreement. https://assets.documentcloud.org/documents/2646274/ Updated-109r01.pdf (Accessed 30 Nov 2016).
- Vaissière, A.C., Levrel, H., Hily, C., Le Guyader, D., 2013. Selecting ecological indicators to compare maintenance costs related to the compensation of damaged ecosystem services. Ecol. Indic. 29, 255–269.
- Van Aalst, M.K., Cannon, T., Burton, I., 2008. Community level adaptation to climate change: the potential role of participatory community risk assessment. Glob. Environ. Chang. 18 (1), 165–179.
 Van Houtven, G., Mansfield, C., Phaneuf, D.J., von Haefen, R., Milstead, B., Kenney, M.A.,
- Van Houtven, G., Mansfield, C., Phaneuf, D.J., von Haefen, R., Milstead, B., Kenney, M.A., Reckhow, K.H., 2014. Combining expert elicitation and stated preference methods to value ecosystem services from improved lake water guality. Ecol. Econ. 99, 40–52.
- Verdolini, E., Anadon, L.D., Lu, J., Nemet, G.F., 2015. The effects of expert selection, elicitation design, and R&D assumptions on experts' estimates of the future costs of photovoltaics. Energ Policy 80, 233–243.
- Vermeulen, S.J., Aggarwal, P.K., Ainslie, A., Angelone, C., Campbell, B.M., Challinor, A.J., ... Lau, C., 2012. Options for support to agriculture and food security under climate change. Environ. Sci. Pol. 15 (1), 136–144.
- Vincent, K., 2007. Uncertainty in adaptive capacity and the importance of scale. Glob. Environ. Chang. 17 (1), 12–24.
- Winkler, R.L., 2015. Equal versus differential weighting in combining forecasts. Risk Anal. 35 (1), 16–18.
- World Bank, 2010. The Economics of Adaptation to Climate Change. http://siteresources. worldbank.org/EXTCC/Resources/EACC_FinalSynthesisReport0803_2010.pdf (Accessed 25 Nov 2016).
- Xu, F., Mittelhammer, R.C., Torell, L.A., 1994. Modeling nonnegativity via truncated logistic and normal distributions: an application to ranch land price analysis. J. Agric. Resour. Econ. 19 (1), 102–114.